Author's response to reviews

Title: The protocol and design of a randomised controlled study on training of attention within the first year after acquired brain injury

Authors:

Aniko Bartfai (aniko.bartfai@ki.se)
Gabriela Markovic (gabriela.markovic@ds.se)
Kristina Sargenius Landahl (sargenius-landahl.kristina@ds.se)
Marie-Louise Schult (schultmarie-louise@ds.se)

Version: 2 Date: 20 March 2014

Author's response to reviews:

We have mailed the scanned copies of the two ethical permits for the project and all funding notifications. We are in the process of providing authorized translation for the relevant parts, but we will emphasize that we cannot send any original documents. We are sorry if that would prevent a publication in BMC Neurology, but administrative routines prevent us to do so. We would be happy to supply authorised copies of the documents if needed.

1. The trial registration number is: NCT02091453
2. The name of the ethics committee was added, page 5 lines 24
3. Ethical and Funding Approval Documentation: copies of all ethical approval and funding notifications were sent as email attachments to BMCSeriesEditorial@biomedcentral.com.
4. Funding: As stated in the paper the study is being financed by: ALF, KID, Promobilia, Stroke förbundet. ALF is the joint founding of Stockholm County Council and the Karolinska Institutet for financing clinical research. KID is the founding of the Karolinska Institutet for financing projects for PhD students, Promobilia is an independent foundation financing research related to disability. Stroke förbundet is an association representing the interests of those with stroke. All funding bodies above apply a peer-review system and none of the organisations mentioned above has any commercial interest in the presented study.
5. Study status: the study has started and we are in the inclusion phase, i.e. recruiting patients.
6. Related Articles: there are no other submitted or published articles regarding this study
7. Individual author contributions: we have expanded this section according to the instructions on page 14, lines 3-17.